Author’s response to reviews

Title: Compound heterozygous LPIN2 pathogenic variants in a patient with Majeed syndrome with recurrent fever and severe neutropenia: case report

Authors:

jun liu (ljun2005@163.com)
Xu-Yun Hu (hu_xuyun@126.com)
Zhi-Peng Zhao (bchpeng@foxmail.com)
Ruo-Lan Guo (guoruolan_bch@163.com)
Jun Guo (guojunbjmu@163.com)
Wei Li (liwei@bch.com.cn)
Chan-Juan Hao (hchjhchj@163.com)
Bao-Ping Xu (xubaopingbch@163.com)

Version: 2 Date: 27 Oct 2019

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and consider our manuscript. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

1 - Please upload your revised manuscript in the following section:

Clinical-Molecular Genetics and Cytogenetics

OK

2 - Please confirm that all the consents required were obtained in order for you to include the data in table 1.

OK, and we have marked the references in the main text.
3 - Please confirm whether the parents consented for their information to be included in the family tree in figure 1.

yes, we have signed the consent.

4 - Please remove the duplicate legend information for the figures and table.

OK

5 - At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

OK